Author’s response to reviews

Title: Are people in residential care entitled to receive rehabilitation services following hip fracture? Views of the public from a citizens’ jury

Authors:

Maria Crotty (maria.crotty@flinders.edu.au)
Emmanuel S Gnanamanickam (emmanuel.gnanamanickam@unisa.edu.au)
Ian Cameron (ian.cameron@sydney.edu.au)
Meera Agar (meera.agar@uts.edu.au)
Julie Ratcliffe (julie.ratcliffe@flinders.edu.au)
Kate Laver (kate.laver@flinders.edu.au)

Version: 1 Date: 23 Apr 2020

Author’s response to reviews:

We sincerely thank the Editor for his feedback considering the two conflicting reviews. We have made some additions and edits to the paper based on the feedback of the Editor and Reviewer 1. We have provided further argument for the significance of the jury in response to the feedback from Reviewer 2 although this reviewer did not request any changes to the paper and so no additional edits were made.

All feedback and response is detailed in the 'response to reviewers' document which was uploaded as supplementary material.